Author's response to reviews

Title: Prevalence and Correlation between Depression, Anxiety, and Other Features in Outpatients with COPD in China

Authors:

Peian Lou (loupeian2004@yahoo.com.cn)
Yanan Zhu (yanan01112@163.com)
Peipei Chen (7776290@qq.com)
Pan Zhang (xzzhangpan@126.com)
Jiaxi Yu (zxyjx007@163.com)
Ning Zhang (494174@qq.com)
Na Chen (14781960@qq.com)
Lei Zhang (18905208811@189.cn)
Hongmin Wu (wuhongmin19870512@163.com)
Jing Zhao (354551183@qq.com)

Version: 2 Date: 14 April 2012

Author's response to reviews: see over
Dear editor:

According to your requirements, I have revised my manuscript as follow:

1. The sentence “evaluated effect of depression and anxiety on patient status with COPD in China” have been revised “evaluated correlation between anxiety, depression, and other features on patient status with COPD in China.”

2. The control group and the COPD who were selected, source population, study and target population have been introduced in parts of sampling strategy and study population of methods,

3. The table 1 have been shared in two different tables: table 1and table 2.

4. In results, the paper mentioned “Subjects who were anxious and depressive symptoms related to poorer health outcomes including a higher BODE index, a shorter 6MWD, more dyspnea, and a l higher SGRQ score.” So in conclusions: we said “Patients with COPD who had anxiety and/or depression had a poorer health-related quality of life.”

5. The name of the ethical board and the approval ID have given in manuscript.

6. Table 3 has been recalculated

7. Depression and anxiety have been replaced by depressive symptoms and anxious symptoms.

Sincerely yours,

Peian L. on behalf of the authors.